Author's response to reviews

Title: Higher rate of skin rash in a phase II trial with weekly nanoparticle albumin-bound paclitaxel and cisplatin combination in Chinese breast cancer patients

Authors:

Li Chen Tang (forever_drawer@163.com)
Bi Yun Wang (wangbivun@msn.com)
Si Sun (0356208@fudan.edu.cn)
Jian Zhang (drawer426@hotmail.com)
Zhen Jia (mark_xyw@hotmail.com)
Yun Hua Lu (mark_xyw@163.com)
Gen Hong Di (dgh_2011@yahoo.cn)
Zhi Ming Shao (zhimingshao@yahoo.com)
Xi Chun Hu (xchu2009@hotmail.com)

Version: 5 Date: 19 March 2013

Author's response to reviews: see over
Dear Editor,

It is our pleasure to receive the suggestion from you. Thank you very much for your suggestion. As you mentioned, we have added the authors' contribution to the paper.
Please help to check it.
Thank you again for your help.